Author's response to reviews

Title: Acute paranoid psychosis as sole clinical presentation of hepatic artery thrombosis after living donor liver transplantation

Authors:

Armin D Goralczyk (agoralczyk@gmail.com)
Giuliano Ramadori (gramado@med.uni-goettingen.de)
Heinz Becker (hbecker@chirurgie-goettingen.de)
Aiman Obed (aobed@chirurgie-goettingen.de)
Thomas Lorf (tlorf@med.uni.goettingen.de)

Version: 2 Date: 10 December 2009

Author's response to reviews:

MS: 1999925691294400

Changes made because of EDITORIAL REQUESTS:

- Competing interests section included

Changes made due to referees verdict:

Referee 1: none
Referee 2:

Major points

1. Changes in "Case presentation" 4th paragraph, "Discussion" 4th and 5th paragraph and "Conclusion" 1st paragraph to reflect that graft dysfunction is present and is most likely related to HAT although elevated bilirubin may indicate earlier graft function not related to HAT or psychosis.

2. Changes made in "Discussion" 4th and 5th paragraph to indicate that there were no clear-cut laboratory changes (neither liver function tests, nor ammonia, nor metabolic disorders) that either indicated cause for HAT or cause for psychosis.

Minor points:

The manuscript has been edited to increase quality of written english

Sincerely yours
A. Goralczyk